Author's response to reviews

Title: Probiotics in the treatment of acute rotavirus diarrhoea. Is better a single species product or multiple species product?: A randomized, double-blind, controlled trial using two different probiotic preparations in Bolivian children.

Authors:

Giuseppe Grandy (ggrandy@inta.cl)
Marcos Medina (marcosmebu@gmail.com)
Richard Soria (rsoria@gmail.com)
Carlos Teran (carteran76@hotmail.com)
Magdalena Araya (maraya@inta.cl)

Version: 2 Date: 13 January 2010

Author's response to reviews:

Alfredo Guarino, M.D.
Department of Pediatrics, University Federico II of Naples;
via S. Pansini 5, 80131, Naples, Italy.
Telephone: +39-081-7463501; Telefax +39-081-229 6037.
E-mail jpgn.eu@tin.it
http://jpgn-europe.edmgr.com